Author's response to reviews

Title: Copula based prediction models: an application to an aortic regurgitation study

Authors:

Pranesh Kumar (pkumar@kfshrc.edu.sa)
Mohamed M Shoukri (shoukri@kfshrc.edu.sa)

Version: 6  Date: 23 May 2007

Author's response to reviews: see over
Copula based prediction models: An application to the aortic regurgitation study.
Pranesh Kumar and Mohamed M Shoukri

May 23, 2007

We thank the BMC editorial production team for their suggestions on the formatting changes. Accordingly, we have made the following changes in the revised manuscript:

1) In the manuscript title, ‘An’ is changed to ‘an’.
2) Removed ‘author contributed equally to this work’ from title page.
3) Abstract in the submission system and in the manuscript are the same.
4) Reference numbers are changed to black and underlining removed.
5) ‘Return to citation in text’ from each reference removed.
6) On page 9, website is removed and cited in the main reference list against [14].
7) In competing interests, change is made to ‘The authors declare that they have no competing interests’.
8) Authors’ contributions are rewritten as suggested.
9) Figures 1a and 1b are renamed as Figure 1 and Figure 2.
10) Figure titles and figure numbers are removed from the image files.

Further, we have carefully checked the manuscript for any typing or content errors and have corrected them.